Author's response to reviews

Title: Arabian incense burning: a common trigger for asthma symptoms in Omani children

Authors:

Omar A Al-Rawas (orawas@squ.edu.om)
Abdullah A Al-Maniri (almaniri@gmail.com)
Bazdawi M Al-Riyami (bazdawi@squ.edu.om)

Version: 2 Date: 31 August 2008

Author's response to reviews:

To The Editor,

BMC pulmonary Medicine

Sub: Submission of our manuscript entitled "Arabian incense burning: a common trigger for asthma symptoms in Omani children"

Dear Editor

We would like to submit the above mentioned paper for consideration in your journal. The article raises the issue of burning the Arabian incense and it effects on children respiratory symptoms.

Regards

Corresponding Author:
Dr. Omar A. Al-Rawas, PhD, FRCP
Department of Medicine,
College of Medicine and Health Science, Sultan Qaboos University
P.O. Box: 35, Postal Code 123, Muscat, Sultanate of Oman
Email:orawas@squ.edu.om
Tel: (+968) 24141133; Mobile (+968) 99888777, Fax: (+968) 24141198